Author's response to reviews

Title: Neosalpingostomy for partial hydrosalpinx in a premenarcheal girl: a case report

Authors:

Stjepan Višnjić (stjepan.visnjic@zg.t-com.hr)
Rok Kralj (rok_kraljzg@yahoo.com)
Božidar Župančić (bozidar.zupancic@kdb.hr)

Version: 3
Date: 18 March 2014

Author's response to reviews:

Dear sirs,
the authors believe that the revised article conforms with the demands stated in the reviewer's reports

Reviewer report Ratko Matijević
1. We have supplemented the discussion with the possible causes of the torsion of the fallopian tube and mentioned that the chain of events in our patient is not exactly clear - whether hydrosalpinx was a cause or a consequence of the torsion of the fallopian tube.
2. We have edited the text for it to be acceptable for publication.

Reviewer report Mohamed Fahmy
1. We have changed the title of the case report
2. The histopathological report is highlighted in the revised article
3. We have mentioned that the most common type of approach in this situations is laparoscopy
4. We have expanded the case report with the data that we have in our archives and expanded the discussion with insights from other articles

We have included the ethnicity of the patient in the revised manuscript.
The patient's gender is changed to female.
We have included a competing interests section header.

Kind regards